Author’s response to reviews

Title: Prevalence of sleep-disordered breathing and association with orofacial symptoms among Saudi primary school children

Authors:

Laila Baidas (lailabaidas@gmail.com)

Asma Al-Jobair (aaljobair@ksu.edu.sa)

Huda Al-Kawari (hkuwari@hotmail.com)

Aram Al-Shehri (aramalshehri@yahoo.com)

Sarah Al-Madani (sarahalmadani.90@gmail.com)

Hana Al-Balbessi (hbalbeesi@gmail.com)

Version: 3 Date: 27 Feb 2019

Author’s response to reviews:

1. Ethics Approval and Consent to Participate

Please move this statement from methods to 'Ethics Approval and Consent to Participate'?

Response

I moved the statement from methods to 'Ethics Approval and Consent to Participate'.

2. Change in Authorship

We note that the authors list has changed since original submission of the manuscript (removal of Pedo Cert and additional author Hana Al-Balbessi). In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

Response

No changes were made in authorship since the first manuscript draft. The name of author Hana Balbessi was included with other authors list in the first draft submitted in Oct 2017. However, it was unintentionally dropped from the author names under the main title but not from the author list. Then in the revised version of manuscript in July 2018 and Dec 2018 the name of the author Hana Balbessi was added.
However, as requested from editorial office, the change in authorship form was completed with all author signatures and sent to editorial office.

3. Authors Contributions

Please ensure the initials on the title page match the initials under 'authors contributions'. For example, Huda Al-Kawari should be HAW.

Response

All the initials are corresponding with the names are indicated in the author contribution.

4. Corresponding Author

Please can you also ensure that the email addresses on the title page match the email addresses we have on our system (specifically the corresponding author).

Please update this either on our editorial system or within your manuscript.

Response

The email of corresponding author (Laila Baidas) was changed from lailabaidas@gmail.com to lbaidas@ksu.edu.sa, because my university prefer to use the official email for the publication.

5. Response Letter Editing Certificate & Consent Form

Thank you for providing a Response Letter Editing Certificate & Consent Form. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.

Response

It is removed

6. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response

The manuscript has been English edited many times by Edanz. In the last version I request a native English speaker to revise it. Additionally, the clean version was send as requested.